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Weekly report - 1
● Currently flushing with ECO2 and stability ongoing

● ABS 69 during the week and EP-DT RPC at full efficiency for ABS 69

● Few interruptions of isobutane flow during the week
● Problem with an EV
● Fixed now by replacing EV but better to keep an eye out
● If isobutane goes below 0.4 → automatic gas kill of stability run

● Two source-off scans taken before the start of irradiation (on April 30th)
● 1078 (here) → KODEL-H TOP > 100 uA, removed from next scan
● 1079 (here)

● Three stability scans → ATLAS and KODEL-H not included
● 258 (stopped due to no isobutane, here)
● 259 (stopped due to no isobutane, here)
● 260 (ongoing, here)

http://webdcsecogas.cern.ch/index.php?mod=HVSCAN&p=dqm&id=1078
http://webdcsecogas.cern.ch/index.php?mod=HVSCAN&p=dqm&id=1079
http://webdcsecogas.cern.ch/index.php?mod=STABILITY_GIFPP&p=dqm&id=258
http://webdcsecogas.cern.ch/index.php?mod=STABILITY_GIFPP&p=dqm&id=259
http://webdcsecogas.cern.ch/index.php?mod=STABILITY_GIFPP&p=dqm&id=260


  

Weekly report - 2

HV to WP

ABS 2.2 ABS 69ABS 2.2Source OFF scans



  

Weekly report - 3



  

Source off scan comparison

Before TB
Trip in last HV



  

Source off scan comparison

Before TB After TB

Before TB
After TB



  

Source off scan comparison



  

Source off scan comparison



  

Instructions for next shifter
● Continue ongoing run (260) with EPDT at WP

● At GIF++ weekly meeting (Thu) we will discuss whether we want to still have 69 or if we are 
satisfied 

● If ABS 69 is maintained 
● EP-DT at WP (same config as run 260)

● If ABS 2.2 is put back
● EP-DT at 10.6 kV
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